%% jrnl_ijiss.tex
%% 
%% This is a skeleton file demonstrating the use of ijiss.cls
%% with an International Journal of Information Security Science(IJISS).
%% This template is a modified version of the Turkish Journal of Electrical Engineering {\&} Computer Sciences template.
%%********************************************************************
\documentclass[journal]{ijiss}
%
% If IJISS.cls has not been installed into the LaTeX system files,
% manually specify the path to it like:
% \documentclass[journal]{../sty/IJISS}
\usepackage{amsfonts,amssymb,amsmath,amsgen,amsopn,amsbsy,theorem,graphicx,epsfig}
\usepackage{eufrak,amscd,bezier,latexsym,mathrsfs,eurosym,enumerate}
\usepackage{url}
\usepackage{tikz,xcolor,hyperref}
\usepackage{academicons,xcolor}
\definecolor{orcidlogocol}{HTML}{A6CE39}
\newtheorem{theorem}{Theorem}
\newtheorem{example}{Example}

%\newcommand{\orc}[1]{\href{https://orcid.org/#1}{\textcolor[HTML]{A6CE39}{\aiOrcid}}}
\newcommand{\orc}{\includegraphics[height=\fontcharht\font`A]{orcidlogo}}

% *** GRAPHICS RELATED PACKAGES ***
%
\ifCLASSINFOpdf
  % \usepackage[pdftex]{graphicx}
  % declare the path(s) where your graphic files are
  % \graphicspath{{../pdf/}{../jpeg/}}
  % and their extensions so you won't have to specify these with
  % every instance of \includegraphics
  % \DeclareGraphicsExtensions{.pdf,.jpeg,.png}
\else
  
  \fi

% correct bad hyphenation here
\hyphenation{op-tical net-works semi-conduc-tor}


\begin{document}
%
% paper title
% can use linebreaks \\ within to get better formatting as desired

\title{Manuscript Template: Full Title}

\author{Name Surname$^{1}$ \href{https://orcid.org/XXXX-XXXX-XXXX-XXXX}{\orc} , Name Surname$^{2}$\href{https://orcid.org/XXXX-XXXX-XXXX-XXXX}{\orc}% <-this % stops a space
\address{$^{1}$XYZ, ABC University, City, Country \protect\\
		$^{2}$ ASD University, City, Country \protect\\
Corresponding Author: mail@mail.com \\
%email of the corresponding author
%\bigskip
%ORCID iD: 0000-xxxx-yyyy-zzzz, 0000-xxxx-yyyy-zzzz \\
%ORCID ID of all authors should be given. 
\smallskip 
Research Paper \hspace{0.5cm} Received: dd.mm.202y \hspace{1cm} Revised: dd.mm.202y \hspace{1cm} Accepted: dd.mm.202y
}
%Received: the submission date.

}



% The paper headers
\markboth{INTERNATIONAL JOURNAL OF INFORMATION SECURITY SCIENCE}{N. Surname, Vol.1, No.1, pp.1-21}%



% use for special paper notices
%\IEEEspecialpapernotice{(Invited Paper)}



% for Computer Society papers, we must declare the abstract and index terms
% PRIOR to the title within the \IEEEcompsoctitleabstractindextext IEEEtran
% command as these need to go into the title area created by \maketitle.
\ijisstitleabstractindextext{%
		
\begin{abstract}
This document is a template for use by authors sending manuscripts to the International Journal of Information Security Science. It also contains important information on the paper style. The title of the manuscript must be written in lower case except for the first word and proper nouns. Author names must be given in full, with surnames (family names) all in capitals. Author addresses must be given in English in the following order: Department, Faculty, University, City, Country, with numbers in superscript after each author name to indicate his/her address. Footnotes must not be used for addresses. The corresponding author's email address must be clearly given, marked with an asterisk. The abstract must not be longer than 300 words and must clearly state the study's purpose and results. The manuscript's title and abstract must not contain mathematical formulae. The abstract must not contain any reference citations. The key words must be separated by commas and should not include acronyms.

\end{abstract}
% IEEEtran.cls defaults to using nonbold math in the Abstract.

\begin{ijisskeywords}
information security, instructions for authors, manuscript template
\end{ijisskeywords}
}


% make the title area
\maketitle


% To allow for easy dual compilation without having to reenter the
% abstract/keywords data, the \IEEEcompsoctitleabstractindextext text will
% not be used in maketitle, but will appear (i.e., to be "transported")
% here as \IEEEdisplaynotcompsoctitleabstractindextext when compsoc mode
% is not selected <OR> if conference mode is selected - because compsoc
% conference papers position the abstract like regular (non-compsoc)
% papers do!

% \IEEEdisplaynotcompsoctitleabstractindextext has no effect when using
% compsoc under a non-conference mode.


% For peer review papers, you can put extra information on the cover
% page as needed:
% \ifCLASSOPTIONpeerreview
% \begin{center} \bfseries EDICS Category: 3-BBND \end{center}
% \fi
%
% For peerreview papers, this IEEEtran command inserts a page break and
% creates the second title. It will be ignored for other modes.




\section{Introduction} \label{Intro}
Authors should use this template very carefully when preparing a manuscript for submission to the International Journal of Information Security Science. \textcolor{red}{The manuscript must be written in \LaTeX. Other formats are not accepted. }The text in this template can be replaced by typing or copy/pasting to form the final form of the manuscript. The style of this document should never be changed. Please make sure that, in addition to this template, you carefully read and follow the ``Instructions for Authors'' in the journal web site. Submissions that do not strictly follow this template and/or all instructions for authors are returned back to authors without review.
\subsection{Sections and subsections}

Articles should be divided into sections and subsections. Principal sections should be numbered consecutively (e.g., as 1. Introduction, 2. Materials and methods, etc.), while subsections should be numbered as 1.1., 1.2., etc. Do not use numbers for the Acknowledgments and References sections. This journal is free of charge (for both authors and readers).

The Introduction should argue the case for the study, outlining only essential background, and should not include details of findings or conclusions. It should not be only a review of the subject area, but it should also contain a clear statement of the question being addressed. In the Materials and methods section, please provide concise but complete information on materials and analytical/statistical procedures used. This part should be as clear as possible to enable other scientists to repeat the research presented. For example, brand names and company locations should be supplied for all mentioned equipment, instruments, chemicals, etc. Results of the study should be given in the Results section. In the Discussion section, where findings are discussed, statements from the Introduction and Results sections should not be repeated unnecessarily. It is advised that the final paragraph of the Discussion section highlights the main conclusions of the study. The Results and Discussion sections may be combined. Where necessary, you can refer to sections like Section~\ref{Intro}.

\begin{figure}[h!]
	\begin{center}
		\includegraphics[width=9.0cm]{figure_example}
		\caption{A sample figure.}
		\label{fig1}
	\end{center}
\end{figure}

Manuscripts must be written in English. Contributors who are not native English speakers are strongly advised to ensure that a colleague fluent in the English language or a professional language editor has reviewed their manuscript. Concise English without jargon should be used. Repetitive use of long sentences and passive voices should be avoided. It is strongly recommended that the text be run through computer spelling and grammar programs. Either British or American spelling is acceptable but it must be consistent throughout. Manuscripts written in poor English will not be submitted to referees for evaluation and will be returned back to authors.  

\textbf{We emphasize that the style of this document, including line spacing, margins, fonts, and page numbers, should never be changed.}  

\subsection{Symbols, units, and abbreviations}

In general, this journal follows the conventions of \textit{Scientific Style and Format, The CSE Manual for Authors, Editors, and Publishers, Council of Science Editors, Reston, VA, USA}. All symbols, such as $\times, \mu, \eta$, and $\nu$, should be written by using corresponding \LaTeX definitions. For example, the degree symbol should be used as $^{\circ}$ and it should not be written by using superscripted letter o or number 0. Similarly, the multiplication symbol must be used as $\times$, not the letter x. Spaces must be inserted between numbers and units (e.g., $3$~kg, $2$~m, $10$~V) and between numbers and mathematical symbols (e.g., $+$, $-$, $\times$, $=$, $<$, $>$), but not between numbers and percent symbols (e.g., 45{\%} is the correct version). Please use SI units. All abbreviations and acronyms should be defined at the first mention. Latin terms such as et al., in vitro, and in situ, should not be italicized. 

\begin{table}[h!]
	\caption{A sample table including some styles.}
	\begin{center}
		\begin{tabular}{|l|l|l|l|}
			\hline
	\multicolumn{1}{|c|}{\begin{tabular}[c]{@{}c@{}}GKA - protocol steps\\ with contributions\end{tabular}} & \multicolumn{2}{c|}{\begin{tabular}[c]{@{}c@{}}Computational \\ cost (ms)\end{tabular}} & \multicolumn{1}{c|}{\begin{tabular}[c]{@{}c@{}}Communication \\ cost (bits)\end{tabular}} \\
	
			\hline
			Round 1 - Step 1& 
			28& 
			${\Theta }(\, 1\, )$& 
			$n\times \, 2048$ \\
			\hline
			Round 2 - Step 2& 
			$13+5n$& 
			${\Theta }(\, n\, )$& 
			$n\times \, 2048$ \\
			\hline
			Round 2 - Step 3& 
			6& 
			${\Theta }(\, 1\, )$& 
			-- \\
			\hline
			Total cost of protocol& 
			$(42+5n)$& 
			${\Theta }(\, n\, )$& 
			${n\times \, }4  \to  {\Theta }(\, n\, )$ \\
			\hline
		\end{tabular}
		\label{tab1}
	\end{center}
\end{table}


 

\subsection{References within the Text and Reference List}

References should be cited in the text by numbers in square brackets. Please do not use individual sets of square brackets for citation numbers that appear together, e.g., use \cite{menezes1997handbook,kirby_2013} or \cite{menezes1997handbook,haynes_2009,abramowitz_64}, not \cite{menezes1997handbook},\cite{haynes_2009},\cite{abramowitz_64}, or \cite{kirby_2013}--\cite{li_2021}. All references cited in the manuscript must appear in the list of references at the end, while all references listed in the reference list must be cited in the manuscript.

References should be formatted as follows (please note the punctuation and capitalization):

\textbf{Book}

Basic Format:
J. K. Author, “Title of chapter in the book,” in Title of His Published Book, xth ed. City of Publisher, (only
U.S. State), Country: Abbrev. of Publisher, year, ch. x, sec. x, pp. xxx–xxx.

Book with chapter title

Basic Format:
J. K. Author, “Title of chapter in the book,” in Title of Published Book, X. Editor, Ed., City of Publisher, State (only U.S.), Country: Abbrev. of Publisher, year, ch. x, sec. x, pp. xxx–xxx.

Examples: 

\cite{menezes1997handbook}  A. J. Menezes, P. C. van Oorschot, and S. A. Vanstone. Handbook of Applied Cryptography. CRC Press, 1997. 

\cite{haynes_2009} P. Haynes, “Al-Qaeda, oil dependence, and U.S. foreign policy,” in Energy Security and Global Politics: The Militarization of Resource Management, D. Moran and J. A. Russell, Eds. New York, NY, USA: Routledge, 2009, pp. 62–77.

\cite{abramowitz_64} M. Abramowitz and I. A. Stegun, Eds.,
Handbook of Mathematical Functions, ser. Applied Mathematics Series 55. Washington, DC, USA: NBS, 1964, pp. 32–33. 

\textbf{Conference Paper}

Conference Paper (Paper Presented at a Conference)

Basic Format:
J. K. Author, “Title of paper,” presented at the Abbreviated Name of Conf., City of Conf., Abbrev. State, Country, Month and day(s), year, Paper number.

Conference Proceedings With Editors

Basic Format:
J. K. Author, “Title of paper,” in Abbreviated Name of Conf., X. Editor, Ed. (location of conference is optional), year, pp. xxx-xxx.

Conference Proceedings With Series Title, Volume Title, and Edition

Basic Format:
J. K. Author, “Title of paper,” in Abbreviated Name of Conf. in Volume Title, in Series Title, ed., year, pp. xxxxxx.

Examples: 

\cite{kirby_2013} K. Kirby and J. Stratton, “Van Allen probes: Successful launch campaign and early operations exploring earth’s radiation belts,” presented at the IEEE Aerosp. Conf., Big Sky, MT, USA, 2013.

\cite{li_2021} X. Liu, M. Wang, “QCCA-Secure Generic Key Encapsulation Mechanism with Tighter Security in the Quantum Random Oracle Model,” in Public-Key Cryptography – PKC 2021. PKC 2021. Lecture Notes in Computer Science, J.A. Garay, Ed., vol. 12710. Cham: Springer International Publishing, 2021, pp. 3-26. 


\cite{amador_2006} A. Amador-Perez and R. A. Rodriguez-Solis, “Analysis of a CPW-fed annular slot ring antenna using DOE,” in Proc. IEEE Antennas Propag. Soc. Int. Symp., in Slot Ring Antennas II, vol. 3, 2nd ed., Jul. 2006, pp. 4301–4304.

\textbf{Journal articles:} Journal titles should not be abbreviated; the whole name of the journal should be given. 

\cite{sanico_2018} G. Sanico and M. Kakinaka, “Terrorism and deterrence policy with transnational support,” Def. \& Peace Econ., vol. 19, no. 2, pp. 153–167, 2018 [Online]. Available: https://doi.org/10.1080/10242690701505419

\cite{wang_2014} W. Q. Wang and H. Shao, “High altitude platform multichannel SAR for wide-area and staring imaging,” IEEE Aerosp. and Electron. Syst. Mag., vol. 29, no. 5, pp. 12–17, 2014.

\textbf{M.Theses (B.S., M.S.) and Dissertations (Ph.D.)}

Basic Format:
J. K. Author, “Title of thesis,” M.S. thesis, Abbrev. Dept., Abbrev. Univ., City of Univ., Abbrev. State, year.

J. K. Author, “Title of dissertation,” Ph.D. dissertation, Abbrev. Dept., Abbrev. Univ., City of Univ.,
Abbrev. State, year.

Examples: 

\cite{Moon_2009}  T. D. Moon, “Rising dragon: Infrastructure development and Chinese influence in Vietnam,” M.S. thesis, Dept. of Natl. Sec. Aff., NPS, Monterey, CA, USA, 2009 [Online]. Available: https://calhoun.nps.edu/handle/10945/4694

\cite{rivera_2010} J. Rivera, “Software system architecture modeling methodology for naval gun weapon systems,” Ph.D. dissertation, Dept. of Comp. Sci., Harvard Univ., Cambridge, MA, USA, 2010.

\textbf{Websites:}

First Name Initial(s) Last Name. “Page Title.” Website Title. Web Address (retrieved Date Accessed).

\cite{Smith_2009} J. Smith. “Obama inaugurated as President.” CNN.com.
\url{http://www.cnn.com/POLITICS/01/21/obama_inaugurated/index.html} (accessed Feb. 1, 2009).

\cite{dep_defense_2017} Department of Defense, “About the Department of Defense (DOD).” Accessed April 18, 2017 [Online]. Available: https://www.defense.gov/About/

\section{Orcid Logo}
\textcolor{red}{In order to compile your document ijiss\_main.tex without errors and to get the orcid logo under the title, make sure that the orcidlogo.pdf is in the folder you are working in. Note that orcidlogo.pdf, ijiss\_main.text, ijiss.cls, references.bib and your supplementary files like figures must be in the same folder.}

\section{Environments}
Different environments can be used in the manuscript. Some examples (theorem, proof, example) are as follows.

\begin{theorem} \label{the1}
	The following simplifying assumptions were made to derive the dynamic model of the PMSM in the rotor reference frame.
\end{theorem}

\begin{proof}
	The following simplifying assumptions were made to derive the dynamic model of the PMSM in the rotor reference frame.
\end{proof}

\begin{example}\label{exm1}
	The following simplifying assumptions were made to derive the dynamic model of the PMSM in the rotor reference frame.
\end{example}

The format of an equation should be like
\begin{equation}
	\label{eq1}
	x+y = 1.
\end{equation}
Then, it can be referred in the text as (\ref{eq1}). If it is short, an equation can be written in line as $x+y=1$.


\subsection{Tables and figures}
All illustrations (photographs, drawings, graphs, etc.), not including tables, must be labelled ``Figure''. Figures must be submitted \textbf{both in the manuscript and as separate files} (see below for the accepted format). Each figure or table must have a caption or a legend with a number (e.g., Table~\ref{tab1}, Figure~\ref{fig1}), unless there is only one table or figure, in which case it should be labelled ``Table'' or ``Figure'' without numbering. Captions must be written in sentence case (e.g., ``Macroscopic appearance of the samples.''). The font used inside the figures should be Times New Roman. If symbols such as $\times, \mu, \eta$, or $\nu$ are used, they should be added as symbol.

All figures and tables must be numbered consecutively as they are referred to in the text. Please refer to tables and figures with capitalization and as unabbreviated (e.g., ``As shown in Figure~2...'', and not ``Fig.~2'' or ``figure~2''). The figures and tables themselves may be given at the end of the text or at appropriate places in the text. \textbf{The resolution of images should not be less than 118 pixels/cm when width is set to 16 cm. Images must be scanned at 1200 dpi resolution.} Scanned or photocopied graphs and diagrams are not accepted. Charts must be prepared in two dimensions unless required by the data used. Charts unnecessarily prepared in three dimensions are not accepted.

\textbf{Figures that are charts, diagrams, or drawings must be submitted in a modifiable format, i.e., our graphics personnel should be able to modify them. All figures must be only in *.pdf format; other formats (e.g., *.eps, *.jpeg, *.png, *.tiff) are NOT accepted. Tables and figures, including caption, title, column heads, and footnotes, must not exceed 16$\times$ 20~cm and should be no smaller than 8 cm in width.} 

The same data or information given in a table must not be repeated in a figure and vice versa. It is not acceptable to repeat extensively the numbers from tables in the text or to give lengthy explanations of tables or figures.

\section{Information about references}
Do not include personal communications, unpublished data, websites, or other unpublished materials as references, although such material may be inserted (in parentheses) in the text. If the author of a reference is an organization or corporation, use its name in the reference list (using an abbreviation in the citation, if appropriate); do not use ``Anonymous''. In case of publications in languages other than English, the published English title should be provided if one exists, with an annotation such as ``(article in Turkish with an abstract in English)''. If the publication was not published with an English title, provide the original title only; do not provide a self-translation.

All authors should be included in reference lists unless there are 6 or more, in which case only the first 5 should be given, followed by ``et al.''. The manuscript should be checked carefully to ensure that the spellings of the authors' names and the years are exactly the same in the text as given in the reference list. References should be formatted as shown at the end of this document. Journal titles should be abbreviated according to Thomson Reuters Web of Science~\textcopyright~abbreviations.

\section*{Acknowledgment}
Acknowledgment and/or disclaimers should appear before References. Names of funding organizations should be written in full. \textbf{The authors are advised to describe author contributions, e.g., as follows: S.A. gave the idea, F.A. did the experiments, F.A. and S.A. interpreted the results, S.A. wrote the paper.} 


\section{Conclusion}
The conclusion is here.





% if have a single appendix:
%\appendix[Proof of the Zonklar Equations]
% or
%\appendix  % for no appendix heading
% do not use \section anymore after \appendix, only \section*
% is possibly needed

% use appendices with more than one appendix
% then use \section to start each appendix
% you must declare a \section before using any
% \subsection or using \label (\appendices by itself
% starts a section numbered zero.)
%


\appendices
\section{Appendix 1}
Appendix one text is here.

% you can choose not to have a title for an appendix
% if you want by leaving the argument blank


% use section* for acknowledgement

\section*{Acknowledgments}
The authors would like to thank.


% Can use something like this to put references on a page
% by themselves when using endfloat and the captionsoff option.
\ifCLASSOPTIONcaptionsoff
  \newpage
\fi



% trigger a \newpage just before the given reference
% number - used to balance the columns on the last page
% adjust value as needed - may need to be readjusted if
% the document is modified later
%\IEEEtriggeratref{8}
% The "triggered" command can be changed if desired:
%\IEEEtriggercmd{\enlargethispage{-5in}}

% references section

\bibliographystyle{IEEEtran}
\bibliography{ieeeabrv,references}

% can use a bibliography generated by BibTeX as a .bbl file
% BibTeX documentation can be easily obtained at:
% http://www.ctan.org/tex-archive/biblio/bibtex/contrib/doc/
% The IEEEtran BibTeX style support page is at:
% http://www.michaelshell.org/tex/ieeetran/bibtex/
%\bibliographystyle{IEEEtran}
% argument is your BibTeX string definitions and bibliography database(s)
%\bibliography{IEEEabrv,../bib/paper}
%
% <OR> manually copy in the resultant .bbl file
% set second argument of \begin to the number of references
% (used to reserve space for the reference number labels box)
%\begin{thebibliography}{1}
%
%\bibitem{NYPZ2011} 
%L. Nataraj, V. Yegneswaran, P. Porras, and J. Zhang, \emph{A comparative assessment of malware classification using binary texture analysis and dynamic analysis}, In Proceedings of the 4th ACM workshop on Security and artificial intelligence (AISec '11), Association for Computing Machinery, New York, NY, USA, pp. 21–30, 2011. DOI:https://doi.org/10.1145/2046684.2046689.
%
%\bibitem{NQZ2018}
%S. Ni, Q. Qian and R. Zhang, \emph{Malware identification using visualization images and deep learning}, Computers \& Security, vol. 77, pp. 871-885, 2018.
%
%\bibitem{KeccakSub3} G. Bertoni, J. Daemen, M. Peeters and G. V. Assche, \emph{The Keccak SHA-3 Submission. Submission to NIST (Round 3 3) 2011}, http://keccak.noekeon.org/Keccak-submission-3.pdf. Accessed: September 23, 2020.
%
%
%
%
%
%\end{thebibliography}

% biography section
% 
% If you have an EPS/PDF photo (graphicx package needed) extra braces are
% needed around the contents of the optional argument to biography to prevent
% the LaTeX parser from getting confused when it sees the complicated
% \includegraphics command within an optional argument. (You could create
% your own custom macro containing the \includegraphics command to make things
% simpler here.)
%\begin{biography}[{\includegraphics[width=1in,height=1.25in,clip,keepaspectratio]{mshell}}]{Michael Shell}
% or if you just want to reserve a space for a photo:

% You can push biographies down or up by placing
% a \vfill before or after them. The appropriate
% use of \vfill depends on what kind of text is
% on the last page and whether or not the columns
% are being equalized.

%\vfill

% Can be used to pull up biographies so that the bottom of the last one
% is flush with the other column.
%\enlargethispage{-5in}



% that's all folks
\end{document}


